

Dear Authors, Your effort to better deepen wrist injuries in professional athletes is really commendable. Moreover, Your paper seems interesting but some concerns have to be better addressed. Firstly, you should describe in a specific section or paragraph the methodology you used for this review. How did you organize this research? Which sources did you use? Which criteria did you use? Which is the Prospero approval code? and so on. Also in the case of a narrative review, you should anyway better clarify materials and methods. Then, all the sections are well structured, but the epidemiology section should be integrated. Please provide an overview of the sports in which upper limbs injuries are frequent, so you can explain the importance of your topic. To do that, I suggest the following references: -Fari, G., Notarnicola, A., Di Paolo, S., Covelli, I., & Moretti, B. (2021). Epidemiology of injuries in water board sports: trauma versus overuse injury. *The Journal of sports medicine and physical fitness*, 61(5), 707-711. <https://doi.org/10.23736/S0022-4707.20.11379-3> -Fari, G., Fischetti, F., Zonno, A., Marra, F., Maglie, A., Bianchi, F. P., Messina, G., Ranieri, M., & Megna, M. (2021). Musculoskeletal Pain in Gymnasts: A Retrospective Analysis on a Cohort of Professional Athletes. *International journal of environmental research and public health*, 18(10), 5460. <https://doi.org/10.3390/ijerph18105460> -Fari, G., Santagati, D., Macchiarola, D., Ricci, V., Di Paolo, S., Caforio, L., Invernizzi, M., Notarnicola, A., Megna, M., & Ranieri, M. (2022). Musculoskeletal pain related to surfing practice: Which role for sports rehabilitation strategies? A cross-sectional study. *Journal of back and musculoskeletal rehabilitation*, 35(4), 911-917. <https://doi.org/10.3233/BMR-210191> -Fari, G., Macchiarola, D., Quarta, F., Bianchi, F. P., Caforio, L., Cutti, A. G., . . . Ranieri, M. (2023). Epidemiology of injuries in wheelchair basketball: Trauma versus overuse mechanism and prevention rehabilitative strategies. *Journal of Prosthetics and Orthotics*, 35(1), 32-37. doi:10.1097/JPO.0000000000000458 Best regards and good luck

Dear Editors, Dear Reviewers, thanks a lot for your time and consideration of our submitted paper. We much appreciated your effort in reviewing our article and your comments. We added a material and methods section in the manuscript as suggested, with all the relevant information. We also modified the epidemiology section by adding the suggested relevant information and the references. We then modified the reference order of the entire manuscript. We thank the reviewer for their positive comments and hope our article will be taken into account for publication. Please do not hesitate to contact us if any issues. Kind regards Dear Editors, thank you very much for your email and hard work on this submission. I uploaded a file with replies to reviewer, I have also copied and pasted my point by point reply to reviewer above. There is no reply to last round reviewer process as no specific comment was made. I have modified the references according to your suggestions and Journal guidelines. I am attaching a revised file with the correct references.



JOURNAL EDITORIAL BOARD'S REVIEW REPORT

Name of journal: World Journal of Orthopedics

Manuscript NO: 89114

Title: Review and update on the management of triangular fibrocartilage complex injuries in professional athletes

Journal Editor-in-Chief/Associate Editor/Editorial Board Member: Karl-Heinz Widmer

Country/Territory: Switzerland

Editorial Director: Jia-Ru Fan

Date accepted review: 2023-12-06 23:12

Date reviewed: 2023-12-06 23:17

Review time: 1 Hour

SCIENTIFIC QUALITY	LANGUAGE QUALITY	CONCLUSION
<input type="checkbox"/> Grade A: Excellent	<input type="checkbox"/> Grade A: Priority publishing	<input checked="" type="checkbox"/> Accept
<input checked="" type="checkbox"/> Grade B: Very good	<input checked="" type="checkbox"/> Grade B: Minor language polishing	<input type="checkbox"/> High priority for publication
<input type="checkbox"/> Grade C: Good	<input type="checkbox"/> Grade C: A great deal of	<input type="checkbox"/> Rejection
<input type="checkbox"/> Grade D: Fair	language polishing	<input type="checkbox"/> Minor revision
<input type="checkbox"/> Grade E: Poor	<input type="checkbox"/> Grade D: Rejected	<input type="checkbox"/> Major revision

JOURNAL EDITORIAL BOARD COMMENTS TO AUTHORS

Dear Authors, your publication provides a good overview about this frequent injury addressing classification and consequent treatment in particular. I did find some minor errors, please see my comments. In conclusion, this manuscript is ready for publication. Kind regards.

EMBASE database were utilized (Please provide search keywords). All Authors independently performed the review and all included articles were scrutinized. Finally included articles and results were merged after common agreement among Authors. Authors included only articles specifically based on the diagnosis and management of TFCC in professional athletes published in the last 20 years. Only level of evidence I-II-III-IV articles were included. Informed consents and ethical



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-399-1568
E-mail: office@baishideng.com
https://www.wjgnet.com

approval were not necessary (narrative review).

LITERATURE SEARCH

A narrative review on published papers on the diagnosis and management of TFCC injuries in professional athletes was performed. PubMed, MEDLINE, Cochrane and EMBASE database were utilized. Search keywords were: TFCC injury; triangular fibrocartilage complex injury; TFCC athletes; triangular fibrocartilage complex athletes; TFCC sport; triangular fibrocartilage complex sport; TFCC professional athletes; triangular fibrocartilage complex professional athletes; TFCC treatments; triangular fibrocartilage complex treatments.